\documentclass[12pt]{article}
\usepackage{graphicx}
\usepackage{amssymb}

\textwidth = 6.5 in
\textheight = 9 in
\oddsidemargin = 0.0 in
\evensidemargin = 0.0 in
\topmargin =0 in
\headheight = 0.0 in
\headsep = 0.0 in
\parskip = 0.2in
\parindent = 0.0in

\begin{document}



\LARGE 

{\hspace{50mm}Quiz 3
\hspace{50mm}Name:\hspace{20mm}}


\large

Directions:  Answer each question to the best of your ability.  You may NOT use a calculator.  Indicate for each question which method of integration you are using and clearly identify how you apply that method in the problem.  Each question is worth 5 points.
\small
\begin{enumerate}
\item $\int \, \frac{\cos x}{\sin x}\,dx$ 
\vfill

\item $\int \, x^2 \cos (x^3-1)\, dx$  

\vfill
\item $\int \,x^2 \ln x\, dx$
\vfill
\vfill
\newpage
\item $ \int \, \frac{5x+7}{x^2+3x+2}\, dx$
\vfill
\item $ \int \, \frac{x+4}{x^2+8x+7}\, dx$
\vfill

\end{enumerate}
\end{document}